Author's response to reviews

Title: Variations in suicide method and occurrence by season and day of the week in Russia and the Nenets Autonomous Okrug, Northwestern Russia: a retrospective population-based mortality study

Authors:

Yury A Sumarokov (sioury@mail.ru)
Tormod Brenn (tormod.brenn@uit.no)
Alexander V Kudryavtsev (ispha09@gmail.com)
Odd Nilssen (odd.nilssen@uit.no)

Version: 3
Date: 8 June 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for the acceptance of our paper “Variations in suicide method and occurrence by season and day of the week in Russia and the Nenets Autonomous Okrug, Northwestern Russia: a retrospective population-based mortality study”. I appreciated the constructive criticisms useful suggestions on its quality improvement.

Let me hereby resubmit our article. All the suggestions and remarks were taken in account in the revised version.

1. The additional files in the submission system were removed.

2. We have performed additional professional language editing before submission. This was done in partnership with “Professional Standards Editing Inc” (USA). We have made some changes in the style that we hope will make them substantial enough to work for the better language quality.

3. The web link for Figure 1. “Map of the Nenets Autonomous Okrug” Available at: http://www.grida.no/graphicslib/detail/nenets-autonomous-okrug-reference-map_4eea shows the appointment for open-access use: “This graphic may be reproduced in any form of educational or non-profit purposes without special permission from the GRID-Arendal, provided acknowledgement of the source is made.”

We really sure that the paper will be of interest to the readers of BMC Psychiatry.

Sincerely,
Yury Sumarokov

Department of Community Medicine,
UiT The Arctic University of Norway,
9037, Tromsø, Norway

Arkhangelsk International School of Public Health,
Northern State Medical University,
Troitsky Av. 51, office 2231, Arkhangelsk 163001, Russia
Phone: +7 (8182) 285759
Fax: +7 (8182) 263226
Mobile: +7 921 2445216
E-mail: ysu002@post.uit.no
sioury@mail.ru